Author's response to reviews

Title: Medicinal Herb Use Among Asthmatic Patients attending a Specialty Care Facility in Trinidad.

Authors:

Yuri N Clement (yuriclem@yahoo.com)
Arlene F Williams (arlene_williams@excite.com)
Derick Aranda (elitescholar06@yahoo.com)
Ronald Chase (rkchase_00@yahoo.com)
Nadya Watson (saltprune7@hotmail.com)
Rochelle Mohammed (rochellemohammed@hotmail.com)
Odia Stubbs (odia_stubbs18@yahoo.com)
Deneil Williamson (devette2@yahoo.com)

Version: 3 Date: 11 January 2005

Author's response to reviews:

The following formatting changes have been made:
1. The list and order of authors in the manuscript are identical to that in the submission system.
2. 'advise' was changed to advice in the abstract and a spellcheck was done on the manuscript.
3. Sentence case is used for all headings and sub-headings.
4. The keywords Section has been deleted.
5. The sentence "The authors declare that they have no competing interests" has been added to the Competing Interests Section.
6. Authors' contribution has been changed to Authors' contributions.
7. Authors' initials are used in the Authors' contributions Section.